Author's response to reviews

Title: Association between polymorphisms in RMI1, TOP3A, and BLM and risk of cancer: a case-control study

Authors:

Karin Broberg (karin.broberg@med.lu.se)
Elizabeth Huynh (liseo_82@hotmail.com)
Karin Schläwicke Engström (karin.engstrom@med.lu.se)
Jonas Björk (jonas.bjork@skane.se)
Maria Albin (maria.albin@med.lu.se)
Christian Ingvar (christian.ingvar@med.lu.se)
Håkan Olsson (hakan.olsson@med.lu.se)
Mattias Höglund (mattias.hoglund@med.lu.se)

Version: 6 Date: 6 May 2009

Author's response to reviews:

In table 1, I have changed the figures (131 instead of 189) in the last column so that they match the manuscript text and the abstract. This does not change the content, the data analysis, or the conclusions of the manuscript.